Author's response to reviews

Title: ROS generation via NOX4 and its utility in the cytological diagnosis of urothelial carcinoma of the urinary bladder

Authors:

Keiji Shimada (keijishi@naramed-u.ac.jp)
Tomomi Fujii (2byori@naramed-u.ac.jp)
Satoshi Anai (sanai@naramed-u.ac.jp)
Kiyohide Fujiimoto (kiyokun@naramed-u.ac.jp)
Noboru Konishi (nkonishi@naramed-u.ac.jp)

Version: 4 Date: 1 October 2011

Author's response to reviews: see over
Dear Sir

RE: “
"ROS generation via NOX4 and its utility in the cytological diagnosis of urothelial carcinoma of the urinary bladder” by Shimada et al. (MS ID: 2135278994569578)

Thank you for the review of our manuscript. We would like to resubmit the revised manuscript according to your suggestion.
Responses to the reviewers’ comments are as follows.

Major revision:

1) Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order.

According to the comments, we have corrected author list. All authors are Japanese.

2) Affiliations: Each affiliation should be written in full, in the following format:
Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

According to the comments, we have corrected authors’ affiliations.

3) Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods, Results, Conclusions

According to the comments, we have corrected Abstract in the revised manuscript.

4) Figures: All figures must be cited; figures not cited will be excluded from the final manuscript. Please do not use ranges when listing figures. Please list all the figures, for example: Figures 5, 6, 7, 8 and 9.

Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not
include vertical lines in your tables.

Tables: Every table provided at the end of the manuscript file must be referred to in the text (currently, table 5 is not). Please do not use ranges when listing tables.

We have corrected Figures and Tables in the revised manuscript.

Minor revisions

We have changed according to the comments except for references. Could you please change the references?

We hope that these changes are acceptable and meet your expectations.

Yours sincerely,

Keiji Shimada (1st author)
Department of Pathology
Nara Medical University School of Medicine
840 Shijocho, Kashihara, Nara, 634-8521, Japan
Tel: +81-744-22-3051: Fax: +81-744-23-5687
E-mail: keijiishi@naramed-u.ac.jp